Author's response to reviews

Title: Experimental infection of Balb/c nude mice with Hepatitis E virus

Authors:

huang fen (hxg@sjtu.edu.cn)
zhang wen (z0216wen@sjtu.edu.cn)
gong ga (gongga@sjtu.edu.cn)
yuan cong li (ycl@sjtu.edu.cn)
yan yi jia (y_yan@sjtu.edu.cn)
yang shi xing (yangshixing@sjtu.edu.cn)
cui li (lcui@sjtu.edu.cn)
zhu jian guo (Zhu_jg@sjtu.edu.cn)
yang zhi biao (zbyang@sjtu.edu.cn)
hua xiu guo (hxg@sjtu.edu.cn)

Version: 2 Date: 16 August 2008

Author's response to reviews:

Dear Editor,

We would like to resubmit our revised manuscript for your consideration. We have rewritten the abstract and made lots of revisions in the text. In the “Materials and methods”, we have clarified the ethical position of our investigation. By doing so we believe that the manuscript is substantially improved. We hope that you will now find the manuscript suitable for peer review.

Sincerely yours,

Fen Huang